Author's response to reviews

Title: An outbreak following importation of wild poliovirus in Xinjiang Uyghur Autonomous Region, China, 2011

Authors:

Haibo Wang (hbwang2005@163.com)
Huiming Luo (hbwang2005@126.com)

Version: 2
Date: 17 October 2014

Author's response to reviews:

Format of our manuscript has been revised according to suggestions one by one.